Author’s response to reviews

Title: Adolescents' reproductive health knowledge, choices and factors affecting reproductive health choices: a qualitative study in the West Gonja District in Northern Region, Ghana

Authors:

Joseph Kyilleh (jmkyilleh@gmail.com)
Philip Tabong (philgh2001@yahoo.com)
Benson Konlaan (bkonlaan@uds.edu.gh)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

We wish to thank the reviewer for his comments. We have addressed all comments. Below are point-by-point response to the comments.

Response to reviewer

Comment: Page 5, line 8, it should be “The ecological……”
Response: On page 5, line 18, “the” has been inserted and highlighted.

Comment: Page 13, lines 17-18, this sentence is not clear
Response: Page 13, lines 17-18, the sentence has been revised. It now reads “To community opinion leaders, men who are not biological parents to adolescent females take care of their educational needs with the intention of marrying them in future”

Comment: Page 13, line 31, change “were” to “was”
Response: On page 13, line 31, “were” has been changed to “was” and highlighted.

Comment: Page 13, line 21, it is “investment in”
Response: On page 13 line 21, “on” has been changed to “in” and highlighted.
Comment: Page 21, line 5, insert “a”
Response: On page 22, line 6, “a” has been inserted and highlighted.

Comment: Page 22, line 21, change word “make” to “makes”
Response: On page 22, line 21 the word “make” has been changed to “makes”

Comment: Page 25, line 2, change word “act” to “acts”
Response: On page 25 line 2, the word “act” has been changed to “acts”

Response to editor

Editor Comments:
Comment 1: Please provide email addresses for all authors within the title page of your manuscript
Response: The email addresses of all the authors have been added and highlighted on the cover page

Comment 2: Please rename your 'Data and material availability' section to 'Availability of data and materials'
Response: “Data and material availability” has been changed to “Availability of data and materials”

Thank you